Author’s response to reviews

Title: The Longitudinal Effect of Subclinical Hypothyroidism on Urine Microalbumin-to-Urine Creatinine Ratio in Patients with Type 2 Diabetes Mellitus

Authors:

juan xie (xiejuan778@163.com)
Xiaoqing Wang (1210768550@qq.com)
Yiyuan Zhang (358861513@qq.com)
Hailun Li (120647660@qq.com)
Yong Xu (haeyxy@163.com)
Donghui Zheng (zddwj@126.com)

Version: 3 Date: 08 Jul 2019

Author’s response to reviews:

Dear Editor,

Thank you for your letter and the reviewers’ comments concerning our manuscript entitled “The Longitudinal Effect of Subclinical Hypothyroidism on Urine Microalbumin-to-Urine Creatinine Ratio in Patients with Type 2 Diabetes Mellitus” (ID: BEND-D-18-00384R2). We have studied the comments carefully and made some corrections. The main corrections in the paper and the responses to the reviewer’s comments are as follows:

Comment 1: Funding

In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response 1: Thank you for your suggestions, we have changed it.(Funding section, page 8, line 11-12).
Comment 2: Authors' Contributions

Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response 2: Thank you for your suggestion, we have changed authors’ full name to full initials. (Authors’ Contributions section, page 8, line 14-15).

Comment 3: Formatting--

Please rename your "Introduction" heading to "Background".

Response 3: Thank you for your kind, we have renamed it. (Background section, page 2, line 2).